
From: Lacy, Stuart C (QO Inc.)
Sent: Tue Apr O6'17:46'.2420'10
To: Bennett, Gord (Q0 lnc.)
Subject RE: Macondo Update sam
lmportance: Normal

Cheers Cord, I hear you on the drilling ahead if they see more hi res sands. Will give.rou a buzz tomorrow later
affernoon/earl'r: evening and get the latest...time for been & decent grub at this end...
Stu
---Original l\lessage-----

From: Bennett, Gord (Q0 Inc.)

Sent: Tue 1/612010 9:04 AM

To: Lac_v. Stuart C (QO Inc,)

Subject: RE: Macondo Update 5arn

Stu.
We are still trying to regain circulation.
They are sending off the Paleo folks. I am hearing rve will drill another hundred feet to get rathole for logging or TD
hcrc ifwc arc unablc to drill ahcad. So rvc know whcrc TD is. Until thcy go to drill thc 100'and find out thcrc is morc
sand and then the.v will drill ahead another 100' Wink!l
Gord

---Original N{essage-----
From: Lary, Stuart C (QO Inc.)

Sent: Monday. April 05, 2010 2:06 PM

To: Bennett. Gord (Q0 Inc.)

Subject: RE: Macondo Update 5am
Blimey it just sounds like the window's too narrow. Weird though as you say that the formation had seen 14.83 ppg
while drilling. Good to see Earl's still on the ball. nothing like writing off a gas peak as drill gas after all that's
happened in this well. So we're not going to Kaskida then? Are the-v.* not doing the P & A's as well?
Aryvay hopc thcsc losscs gct sortcd and scc 1'ou Thursday chicfl..

Stu

---Original N{essage---

From: Bennett, Gord (Q0 Inc.)

Sent: Mon 4/5/2010 l:40 PM

To: Lacy. Stuart C (QO Inc.;

Subjcct: RE: Macondo Updatc 5am

Oh pah! Lots of fun. We are at a 14.3 MW and drilled the M56 sand yesterday. Geotap'd it and got a 12.58 ppg
EMW. The previous sand ue Geotap'd rvas 14.2 ppg. When we first started this hole section uith 14.3 and bumped
upto' l4.5onsus;ryctedPPincrease.$'estartedtakinglosses. Sol ighteneduptol4.3andtheLCMseernedtoseal
off the culprit sand.
Nobo$.'knorvs where the problem is and so the engineers are lightening up the mud to 14.0 and going to hit the
upper sand with LCM. See if that works. If it doesn't they rvill try Plan B. C through H.
I knew we were on the edge.vesterday r.vhen we got a ?80 unit connection / pumps offgas and Earl said it was drill
gas. I moaned to Jon that we didn't har.e control of this well Q didnt knorv rvhat was going on but I I --,
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not good. ) and Earl was just going to dnll blindly on. Jon called John. and John and Earl decided to trip for bit and
reamer as ROP dropped to 5 ff/hr. They decided to increase the MW to 14.4 1o keep the same ESD we had while
drilling for the trip. Made sense to me. It was pumped around the bit and ras coming up the annulus, u'hen the hole
r\€nt to lotal loss of return in a few nrinutes. Baffling, as we drilled the sands with l4 83 ECD and that is what rve had
l"ith the 14.4 coming back up the annulus.
Nou they are lightening up to 14.0 as they should hare enough ECD to cover the 14.2 sand. Ailvay. rve will see if
all this gives us a kick.....
Looks like -r'ou will be the designated logger.
Then the rig goes to Nile well according to $hat lye har-e been told here.
Regards.

Gord

----Original Message----

From: Lacy, Stuart C (QO Inc.)

Sent: Monday, April 05. 2010 12:41 PM

To: Bennett, Gord (Q0 Inc.)

Subject: RE: Macondo Update 5an

llaving fin again I see!

---Original Message----

From: Bennetl Gord (Q0 Inc.)

Sent: Mon 4/52010 5:18 AM
To: Albertin. Martin L.; Arca, Serkan; Bellow, Jonatlnn M; Bennett, Cord (QO Inc.); Bodek, Robert: Boesiger. Todd ,
Bondurant. Charles H: Cocales, Brett W: Decalf. Carole: Depret. Pierre-Andre ; Guide, John, Halle. Mark E; Johnston
Paul J (Houston); Lacy, Stuart C (QO Inc.); LeBleu, John B; Lundquist, Jason Jl McAughan, Kelly; Morel, Brian P:
Nguyen, Binh Van: Paine, Kate (QuaDril Energ, LT): Piccoli. Leonardo H: Reiter, Doris: Ritchie, Bryan: Scherschel,
Craig: Simpson. Brad: Sims, David C, Skripnikova, Galina; Vinson. Graham (Pinhv); Zamorouer', Alexander V
Subject: Macondo Update 5am

Status:

Hole Depth:

TVD Depth

Bit depth:

Wcll control aftcr losing frrll rcturns

18,260'md

18,249'rvd

18.217 'md

Comments: Stopped drilling at 18,260' md. Evaluated pumps offgas of 786 units obsen'ed at 17:00. As ROP had
dropped to 4 ft/hr, decision was made to trip for new BHA. Increased MW to 14.4 ppg to 14.3 ppg to keep ESD at
same level while tripping. When the 1,1.4 ppg mud got around the bit and up to 15,000'depth lerel in the annulus,
observed total lost returns. Pumped 171 trbls of 84 ppg LCM pill to rnidrright. No change fur losl retunrc. Do not lnre
an update of the situation from midnight at this point in time.

Gord Bemett

Wellsite Geologist

Macondo MC252 #1

Deepwater Horizon

281-366-7744
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